Author's response to reviews

Title: Pancreatic Cancer and Depression - Myth and Truth

Authors:

Martina Mayr (Martina.Mayr@lrz.tum.de)
Roland M. Schmid (Roland.Schmid@lrz.tum.de)

Version: 2 Date: 8 October 2010

Author's response to reviews: see over
Dear Editors,

Thank you for the opportunity to resubmit our manuscript “Pancreatic Cancer – Myth and Truth”. Please find attached the revised version of the paper which has been copyedited according to the helpful editorial and reviewer comments. We had the paper reviewed by a native English speaker and we highlighted all the changes we made; we also added the acknowledgements. Finally we reassessed the manuscript form according to the journal style. We look forward to hearing from you.

Sincerely yours

Martina Mayr, M.D.

Department of Internal Medicine II
Klinikum rechts der Isar
Technical University of Munich
Ismaninger Straße 22
D-81675 Munich
Germany
Phone: 0049-89-4140-2260
Fax: 0049-89-4140-4964
E-mail: martina.mayr@lrz.tum.de